Author's response to reviews

Title: The inhibitory effect of Mesembryanthemum edule (L.) bolus essential oil on some pathogenic fungal isolates.

Authors:

Beauty E Omoruyi Dr (sayhi2sa@yahoo.com)
Anthony A.J Afolayan Prof (aafolayan@ufh.ac.za)
Graeme Bradley (gbradley@ufh.ac.za)

Version: 3 Date: 27 February 2014

Author's response to reviews: see over
Dear editors

We have made all necessary corrections to our manuscript which was previously titled “Antifungal activities of essential oil of Carpobrotus edulis (L.) Bolus” and has been edited to a new title “The inhibitory effect of Mesembryanthemum edule (L.) bolus essential oil on some pathogenic fungal isolates”. The comments made by the reviewer are highlighted in red colours, while our responses are highlighted blue. We hope to hearing from you soon.

Best regards

Beauty et al.